Author's response to reviews

Title: Spontaneous dural tear leading to intracranial hypotension and tonsillar herniation in Marfan syndrome: a case report

Authors:

Aqueel H Pabaney (ahpabaney@yahoo.com)
Farhan A Mirza (farhan.amirza@gmail.com)
Nadir A Syed (nadir.syed@aku.edu)
Humera Ahsan (humera.ahsan@aku.edu)

Version: 7 Date: 27 June 2010

Author's response to reviews: see over
Dear Editors,

Please find the changes done, as requested.

1. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is ‘First name Initial Last name’ (e.g. Joe F Bloggs). Currently they do not match.

DONE.

2. References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example ‘text [1].’ Currently several references are cited out of numerical order, while several are displayed as superscript numbers.

DONE. Please note that reference # 3 is cited in numerical order in the first paragraph of the background, and out of numerical order at the end of the third paragraph as the fact was cited from the same article, and hence cannot be subjected to the same number series. I hope that it is not an issue.

Sincerely,

Farhan A. Mirza